Author's response to reviews

Title: Making sense of evidence in management decisions - the role of research-based knowledge on innovation adoption and implementation in healthcare. Study Protocol

Authors:

Yiannis Kyratsis (y.kyratsis@imperial.ac.uk)
Raheelah Ahmad (raheelah.ahmad@imperial.ac.uk)
Alison H Holmes (alison.holmes@imperial.ac.uk)

Version: 3 Date: 7 March 2012

Author's response to reviews:

07 March 2012

Dear Professor Eccles

Re: MS: 1285607361662766

Making sense of evidence in management decisions – the role of research-based knowledge on innovation adoption and implementation in healthcare. Study Protocol

We have addressed the comments and accepted all changes in the copy-edited version of our manuscript.

Please find enclosed a clean version of the manuscript.

We look forward to seeing the paper progress to publication.

Yours sincerely

Dr Yiannis Kyratsis, Dr Raheelah Ahmad, Professor Alison Holmes